Dear Editor-in-Chief,

On behalf of all the contributing authors, I would like to express our sincere appreciations of your letter and reviewers’ constructive comments concerning our article entitled “Analysis of Clinicopathological Features and Prognostic Factors of Breast Cancer Brain Metastasis” (Manuscript No: 87590). These comments are all valuable and helpful for improving our article. According to the editor and reviewers’ comments, we have made modifications to our manuscript to make our results convincing. In this revised version, changes to our manuscript were all highlighted within the document by using yellow-colored text. Point-by-point responses to the nice associate editor and two nice reviewers are listed below this letter. The revisions are as follows.

Reviewer 1
1 Minor editing is required for the manuscript.
Thanks for your suggestion about language. We have tried our best to polish the language in the revised manuscript.
2 The limit of the study should be listed and discussed.
P13, Line 373-387, we listed and discussed the limitations of the study.
3 References should be edited according to the journal’s guideline.
We have checked the literature carefully and updated references, edited in accordance with the requirements of the journal’s guideline.

Reviewer 2
This is an interesting study of the clinicopathological features and prognostic factors of breast cancer brain metastasis. The study is well designed and the results are well discussed. The reviewer recommends to accept this study after a minor editing of the manuscript.
After careful proofreading by several of our authors, we found there were different format about Figures, we have made them to uniformity. we added abbreviations used in tables. core tip and introduction, we apologize for the confusion generated by the previous version of the manuscript and sincerely hope that our logic is now easier to follow with this new version.
Finally, I would like to thank you again for your guidance and for reviewing and revising my paper again.
Best regards,
Guo-Li Gu, FRCS (Gen Surg), Chief Doctor, Department of General Surgery, Air Force Medical Center, Chinese People's Liberation Army.
2023.9.29